Author's response to reviews

Title: A retrospective analysis of eleven cases of invasive rhino-orbito-cerebral mucormycosis presented with orbital apex syndrome initially

Authors:

Nan Jiang (yankejiang@126.com)
Guiqiu Zhao (zhaoguiqiu_good@126.com)
Shanshan Yang (sherry1216@163.com)
Jing Lin (yankejinjing@126.com)
Liting Hu (yankehu@126.com)
Chengye Che (yankeche@126.com)
Qian Wang (yankewangqian@126.com)
Qiang Xu (yankexuqiang@126.com)

Version: 7
Date: 7 January 2016

Author's response to reviews:

Jan. 7, 2016
Re: Manuscript 1107153785156928
Dear Iskra Staneva,

On behalf of my co-authors, thank you very much for giving us an opportunity to revise our manuscript again. We have carefully revised the manuscript and added the essential details “consent was obtained to publish these images of the patients” in the section of “Methods” according to your comments. We really hope that the revised manuscript meets the editorial requirement. And we do hope that our manuscript will be accepted for publication in your journal. Here we would like to express our sincere gratitude to you.

Yours sincerely,
Nan Jiang, Guiqiu Zhao, Shanshan Yang, Jing Lin, Liting Hu, Chengye Che,
Qian Wang, Qiang Xu

Department of Ophthalmology,
The Affiliated Hospital of Qingdao University, Qingdao, Shandong Province, China